Author’s response to reviews

Title: Spectrophotometric Analysis Evaluating Apical Microleakage in Retrograde Filling using GIC, MTA and Biodentine: An in-vitro study

Authors:

Manisha Nepal (manisha.nepal.mn@gmail.com)
Snigdha Shubham (drsnigdhashubham15@gmail.com)
Rupam Tripathi (rupam.tripathi121@yahoo.com)
Jwolan Khadka (khadkajwolann@gmail.com)
Deepa Kunwar (drkunwardeepea@gmail.com)
Vanita Gautam (drvanitagautam@yahoo.com)
Narayan Gautam (ng_bp22@yahoo.com)

Version: 3 Date: 24 Jan 2020

Author’s response to reviews:

1) The specific individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section has been included as follows in clean manuscript:

NM, KJ and GN performed laboratory experiments. SS and GN analyzed the data and evaluated the result. NM and SS wrote the manuscript. TR, KD and GV reviewed the manuscript and proof reading. All the authors read and approved the manuscript.

2) rename your ‘Graph 1’ to ‘Figure 1’. It has been amended as Figure 1 in clean manuscript.

3) Clean version of manuscript with necessary correction has been attached.